Dear Editors,

Thank you for reading and commenting our manuscript. Attached you will find our adjustments in the revised version.

Here a summary of the changes made to the manuscript:
- We formulate a new running title “Rendez-vous procedure for complete esophageal obstruction”.
- An audio core tip was added (See attachment)
- All reference numbers across the text are put in square brackets in superscript.
- The ‘comments’ are formulated
- General: Some spelling errors are corrected

All changes are made yellow in the revision manuscript.

This case report was reviewed by our local IRB and according to local guidelines no official review was necessary.

We hope to have you sufficient informed, in case of questions please contact us.

Sincerely,

Marjan Klinkert, MD

On behalf of all authors,
Mechteld de Jong, Meindert Sosef, Annick van Nunen, Henricus Belgers